Author's response to reviews

Title: Does transient cART started during primary HIV infection undermine the long-term immunologic and virologic response on cART resumption?

Authors:

Evguenia Krastinova (krastinova@gmail.com)
Remonie Seng (remonie.seng@inserm.fr)
Jerome Lechenadec (jerome.lechenadec@inserm.fr)
Henri Panjo (henri.panjo@inserm.fr)
Asma Essat (asma.essat@inserm.fr)
Djamila Makhloufi (djamila.makhloufi@chu-lyon.fr)
Martine Obadia (obadia.m@chu-toulouse.fr)
Louis Bernard (l.bernard@chu-tours.fr)
Cecile Goujard (cecile.goujard@bct.aphp.fr)
Laurence Meyer (laurence.meyer@inserm.fr)

Version: 2
Date: 13 October 2014

Author's response to reviews:

Dear Editor,

In this revised version of the manuscript we provided the changes you have asked for.
1. We added line numbering and page numbers.
2. In the abstract section we added “Background” as requested for this section
3. In the manuscript on page 2 we added: “Authors declare no conflict of interest financial or non-financial“
4. We moved the titles of Figure 1 and 2 at the end of the manuscript as requested in the author’s instructions

With best regards,
On behalf of the authors
Dr. Evguenia Krastinova